Vrije Universiteit Amsterdam

Universiteit van Amsterdam





Master Thesis

Infer - A Full Scale B2B Sales Predictor

Author: Kailainathan Muthiah Kasinathan (2662871)

1st supervisor: 2nd reader:

Dr. Adam S.Z. Belloum daily supervisor: Dr. Adam S.Z. Belloum supervisor name

A thesis submitted in fulfillment of the requirements for the joint UvA-VU Master of Science degree in Computer Science

August 17, 2021

"Sometimes, the wrong train takes you to the right station." from Crash Landing on You

Abstract

Sales forecasting is an important step in a Business to Business (B2B) company's financial planning. Sales forecasting helps to estimate the future sales and is mostly based on historical data. Accurate estimation of sales allows a company to find out about the cash movement and also about the variation in demand of the products. The area of Machine Learning provides a way to analyse the historic data from the sales pipeline of the company. Keeping Sales Forecast as the domain, The thesis aims to achieve three goals - Predict whether a sales opportunity will be won or lost, formulate a methodology to get the aggregated forecast values at an Area level from the predicted output and to finally develop visualizations to monitor sales pipeline data's hygiene and health aspects. This is aimed to provide an alternative to conventional forecasting process and also to see how data driven forecasting can be used to challenge the conventional process. CatBoost Classifier predicts the sales opportunities and classifies them as won or lost, and a mathematical approach is used to adjust the aggregated forecast value. Tableau is used to develop dashboards to monitor the sales pipeline's hygiene and health. We obtained acceptable results from this approach when compared to forecasts formulated by the conventional process. This thesis also exposed us to a variety of challenges when we do B2B forecasting with machine learning, which are explained as part of the thesis.

Acknowledgements

I would like to thank Dr.Adam Belloum for assisting me throughout the entire scope of the thesis and also for guiding me in the right direction. I would like to thank my intern manager at NetApp, Mr.Minh HoDac and Senior financial Analyst, Mr.Németh Mate for introducing me into this research area and helping out with the data understanding. I would like to thank all the members from the sales and finance teams of NetApp, who were kind enough to help with answers to my questions related to the thesis. I would also like to thank Mr. Rajesh Gupta for helping me out in brainstorming ideas and taking part in discussions which helped improve the thesis. On a final note, I would like to thank my family for everything.Special thanks to Mr.Asif Hassan, Ms.Diksha Das and Mr.Navin Kumar, who were always a constant support during the entire duration of my masters.

Contents

Li	st of	Figur	es		v
Li	st of	Table	5		vii
G	lossa	ry			ix
1	Intr	oducti	ion		1
2	Bac	kgrou	nd		5
	2.1	Under	standing	B2B Sales Process	 5
		2.1.1	B2B Sal	es Process - Stages	 5
		2.1.2	Opportu	unity Management	 7
			2.1.2.1	Sales Stages	 7
			2.1.2.2	Forecast Category	 8
	2.2	Conve	ntional F	orecasting Process	 8
	2.3	Resear	rch Probl	em and Questions	 9
			2.3.0.1	Research Question - 1	 9
			2.3.0.2	Research Question - 2	 11
			2.3.0.3	Research Question - 3	 11
3	\mathbf{Rel}	ated V	Vork		13
	3.1	Machi	ne Learni	ng and B2B Forecasting	 13
4	Dat	a Ove	rview		17
	4.1	Data 1	Explanati	on	 17
	4.2	Train	and Test	Data	 18
		4.2.1	Train D	ata Profiling	 19
	4.3	Featur	re Engine	ering	 19
		4.3.1	Data Pr	eprocessing	 19

Re	efere	nces	References 65			
9	Con	nclusion	61			
	8.2	Idea - All stage B2B Predictor	59			
	8.1	Limitation	57			
8	Lim	nitations and Future work	57			
	7.3	Data Challenges	55			
	7.2	Data Drift	52			
	7.1	Categorical Encoding Difficulties	51			
7	Cha	allenges of Machine Learning in B2B Sales Forecasting	51			
			- 1			
		6.2.2 Forecast Performance Metrics & Visualizations	47			
	0.2	6.2.1 Classification Metrics	42			
	6.2	Evaluating CatBoost Model Performance - Chosen Performance Metrics	49			
U	њлр 6 1	Experiment	ч 1			
6	Eve	periment & Evaluation of Results	/1			
		5.5.2 Active Flow	39			
		5.5.1 Training Flow	39			
	5.5	Model Flow	38			
		5.4.0.3 Pipeline Health - Active / Past	37			
		5.4.0.2 Pipeline Hygiene - Active	37			
		5.4.0.1 Predictor Output Dashboard	37			
	5.4	Visualizations	33			
	5.3	Mathematical Approach	32			
	-	5.2.1 CatBoost	31			
	5.2	Machine Learning Algorithm - Classification Model	30			
0	5.1	Overview	2 9			
5	The	Hybrid Approach - Machine Learning with Simple Maths	29			
	4.4	Feature Selection Approaches	24			
		4.3.4 CatBoost Encoding	23			
		4.3.3 Categorical Encoding	22			
		4.3.2 Adding New Features	20			

List of Figures

2.1	Different Stages and Forecast Categories	7
4.1	Missing Data - Split by Features	20
4.2	Heatmap - Displaying correlation	21
5.1	Design Components	30
5.2	Mathematical Approach to formulate forecast - Steps	34
5.3	Mathematical Approach to formulate forecast - Example	35
5.4	Predictor Architecture	38
5.5	Infer - interface	40
6.1	Class representation of Test Data - Training Flow	43
6.2	Confusion Matrix of Test Data - Training Flow	44
6.3	Quality Metrics by Class for Test Data - Training Flow	45
6.4	ROC Curve of the Test Data - Training Flow	46
6.5	CatBoost Explainability of the Test data - Training Flow	47
6.6	Predictor Output Dashboard	48
6.7	Pipeline Hygiene - Active	49
6.8	Pipeline Health - Past	49
6.9	Pipeline Health - Active	49
7.1	Drift - Sales Area	54
7.2	Drift - Duration	54

List of Tables

4.1	Dataset - Size/Shape	18
4.2	Train/Test - Details $\ldots \ldots \ldots$	19
4.3	New Features Description - Details	21
4.4	CatBoost Encoding Performance	22
4.5	Final list of Features	24
4.6	Feature selection Approaches	27
5.1	Python Libraries	39
6.1	CatBoost - Classification Performance Metrics	45

Glossary

- Account A success full Opportunity which generates revenue for the company.
- Actual Sales A list of all completed opportunities which turned into revenue for the company.
- **B2B Sales** Business to Business. Selling products between two Businesses.
- **B2C Sales** Business to Consumer. Selling products between a business and a con-

sumer mostly an individual.

- **CRM** Customer Relationship Management tool.Tool to administer interactions with customers.
- Lead A customer willing to buy from a company
- **Opportunity** Also referred to as Deal. Provides details about what the customer intends to buy or simply Deal details.
- Sales Health Health is good if there are more potential opportunities present in the sales pipeline that will be completed in the future and vice versa
- Sales Hygiene Ensuring the quality of the data in the Sales Pipeline.
- Sales Pipeline A list of all expected/completed Opportunities.

LIST OF TABLES

1

Introduction

Sales plays a key role in determining the success of an Organization. In layman terms, Sales is selling a product or a service to a customer in return for a remuneration. Business to Business (B2B) Sales is selling directly to other Businesses, whereas Business to Customer (B2C) Sales is selling directly to an individual customer. Organizations can be classified into these two categories based on how they sell their products or solutions. To understand B2B and B2C sales, we provide an example for each of these types. For B2B Sales, consider a Laptop. Suppose a display component is manufactured by a company "A" which is used by the Laptop manufactured by company "B" then A is said to be having a B2B Sales transaction since the sales is between two businesses. On the other hand, e-commerce sites are a very good example for B2C transactions as the sales is between a business and an individual. We only consider Business to Business(B2B) sales in accordance with the thesis. Businesses are often run based on Planning, and It's essential to understand how sales will be in the upcoming year to plan accordingly. This makes forecasting Sales an important component in the Financial Planning process.

Sales forecasting is the process of estimating the future sales based on the current and historical sales transactions. Why is Sales Forecasting important? Sales forecasting is an important jigsaw to the overall puzzle of Financial Planning. The results of Sales Forecasting essentially helps to understand the demand forecasting, cash Flows in an Organization and also helps maintain a better inventory. This relationship of Sales Forecasting also makes it a tough procedure since it needs to be highly accurate to make the dependent processes obtain high quality results. Accuracy of forecast is the primary metric to evaluate the quality of the forecast. An overestimated forecast can lead to higher expectations in sales and also lead to a high inventory, which can cause potential loss to the company.

1. INTRODUCTION

Similarly, an underestimated forecast can cause the company serious troubles when there is a high demand and can also lead to a low inventory.

In a B2B setup, Sales Forecasting differs from the normal ways in which one forecasts sales of a supermarket or a toy store. Everything is based on Opportunities in the B2B setup, and it's easily stored and monitored using the ever improving Customer Relationship Management (CRM) systems. Opportunities are deals which are agreed in principle between the two parties and will be completed in a given time frame.B2B companies generally have an Opportunity Pipeline which will contain all these deals and their details. Traditionally, Sales teams just pick out opportunities from the Pipeline as per their knowledge or experience with the customer and use them to formulate the forecast. Even though this process works, It still does not provide enough evidence to support why a certain Opportunity should be picked or avoided in the forecast process. It's purely based on the individual's inputs, making this process unidirectional and biased based on the individual handling the opportunity. Sometimes this leads to underestimated or overestimated forecast.

With the rise of Machine learning, Systems are becoming intelligent in the way they understand the data. Machine learning opens up an interesting possibility for a system to learn from the data at hand, understand and predict the way the future looks. Forecasting is an interesting area of Application for Machine learning. Although machine learning is widely used in weather forecasting, forecasting the sales using historical selling data etc., It's still not being put in practice in terms of Sales Forecasting in B2B processes where we need to understand the opportunities to predict how the forecast will look like. This thesis aims to solve the problem at hand by building a predictor model that will classify the Opportunities into the two areas, whether it will be completed or not completed. We also try to identify and list out the best Machine learning algorithms which can be used in terms of commercially acceptable accuracy levels. The thesis was carried out at the Worldwide Financial Planning and Analysis team of NetApp - a hybrid cloud data services and data management company. The company's Sales data was used to model the predictor and to evaluate the results of the model being built. Upon completion, the model was deployed to be used as the Forecast predictor for the Worldwide Financial Planning and Analysis team.

This thesis is organized into the following chapters:

• Chapter 2 gives the Background where we discuss in detail about the opportunity cycle lifeline, Traditional approaches etc, and it concludes with drafting out our

research questions and their motivation.

- Chapter 3 explains the related work associated with machine learning in the Sales Forecasting domain.
- Chapter 4 explains the overview about the data, feature engineering approaches and feature selection approaches considered in the thesis.
- Chapter 5 explains the hybrid approach used in the thesis where we discuss the three components of the thesis namely Machine learning algorithm, Mathematical approach and the visualizations. This chapter also explains the architecture of the model Training flow and the active flow.
- Chapter 6 outlines the experiment setup and the evaluation of the results of the three components.
- Chapter 7 discusses the challenges faced by applying machine learning in a B2B sales forecasting setup.
- Chapter 8 briefs on the limitations and the future work associated with the thesis.
- We conclude the thesis report with Chapter 9 which explains the conclusions we obtained from performing this thesis.

1. INTRODUCTION

$\mathbf{2}$

Background

In this chapter, We explain in detail about the Opportunity Life Cycle to provide a better understanding of B2B Sales process. Additionally, we shed some light on the traditional way of formulating forecasts by the Sales Representatives. This chapter concludes by giving the aforementioned explanations and also lists out our research questions which will form the basis for the rest of the paper.

2.1 Understanding B2B Sales Process

With the advancement in technologies, Customer Relationship Management (CRM) Systems have come to the fore of the Sales processes. According to Shoemaker, CRM is a tool that blends all the essential information systems like Sales, Marketing and Service to build partnerships with customers. A recent survey by BuyerZone stated that 91 % of the companies with more than eleven employees now use CRM. CRM's have been adopted by almost all B2B businesses due to the benefits obtained from them stated by Peterson Et Al. like High efficiency, increased productivity of sales processes, better forecasting and performance and ability to document customer needs better. Although CRM is a wider topic area and many research activities have been carried out to identify their benefits, Usage, Implementation, effectiveness etc, we do not deep dive into that. We just establish the definition of CRM to understand the context of the Sales process in the following sections of this chapter.

2.1.1 B2B Sales Process - Stages

The B2B Sales process is similar to the Sales Funnel, and it involves the working of Sales, Marketing and Finance teams. It follows the seven stages involved in the sales funnel,

2. BACKGROUND

which can be broadly placed into three groups, namely -

- Lead Generation
- Opportunity Management
- Completion of Sales

The Lead generation process is the primary step. In this step, Marketing teams reach out to potential customers through different means such as Business meetings, E-mails etc. Once they establish a contact with the customer, they interact with the customers and explain about the products and how it can solve their problems. Upon interest from the customer to buy the product, the team hands over the details of the customer to the Sales teams, thereby marking the customer as a Lead.

Next comes the Opportunity management step. Once the Sales team takes over the handling of the lead, they start understanding the complex asks of the customer and device a quote. This quote is an estimate of the Sale of Products to the customer. The negotiations take place between the Sales team - ideally a Sales Representative and the customer about the quote and the solution offered to them. When both the parties come to a consensus, the deal gets flagged as an opportunity and goes into the Sales pipeline of the company. This opportunity then goes through different stages to reach the Won/Lost stage. The Won/Lost stage is essentially the final step in Opportunity Management post which it moves into the final step in the Sales process. When an opportunity is closed successfully, the customer essentially becomes an Account with the company. In the completion of Sales, both the Sales team and Finance teams play a role. As the opportunity gets completed, It contributes to the revenue of the company where the Finance teams start performing their activities like estimation of revenues, planning for the next Financial year etc. The Sales team also follows up on the order by keeping in touch with the customer for future orders, renewal of product subscriptions, reconfiguring the existing product etc. Where does CRM come into play? CRM contains all the information about the lead, Opportunity and Account and acts as the repository for the Sales information. Lead details are handled by the marketing team whereas the Sales teams constantly update the details about the opportunity such as sales stage, value changes in the opportunity, close date etc. This makes CRM a very handy tool from which a plethora of data can be obtained to perform analysis and also use them to build advanced models and automation's. For the context of the thesis, We look only into the Opportunity Management section of the Sales B2B process and skip the remainder.

Sales Stage	Forecast Category
Prospecting	Commit
Qualification	Best Case
Proposal	Pipeline
Acceptance	Omitted
Negotiation	Closed
Won	
Closed/Lost	

Figure 2.1: Different Stages and Forecast Categories

2.1.2 Opportunity Management

A brief insight into Opportunity Management was given in Subsection 2.1.1. In this part, we try to provide a detailed explanation into the different stages an Opportunity goes through before getting classified as Won or Lost Opportunity. We also outline other details associated to Opportunities in a Sales Pipeline.

Every Opportunity has to go through different Sales Stages and Forecast Categories when it resides in the Sales Pipeline. In general, Sales Stages act as an Opportunity tracker. Sales representatives constantly update this as the Opportunity progresses towards its final stage.Forecast category on the other hand helps to understand the probability that an Opportunity will get completed in the agreed time period. This is also updated by the Sales representatives. The Sales Representatives who own the Opportunity are called Opportunity owner, and they own and update the Opportunity details. Every company has its own way of determining the Sales stages. To maintain the consistency, We will look at the Sales stages and Forecast categories determined at the Company in which the thesis was carried out.

2.1.2.1 Sales Stages

Figure 2.1 shows the different types of Sales Stages. Prospecting stage is the one at which a lead or genuine piece of business is reviewed by the Sales team before moving into the Qualification stage. An opportunity is set to reach the Qualification stage when the team meets with the customer and has qualified that there is a business requirement and a potential budget to solve the need. Post the qualification of the business requirement, a solution is formulated as per the proposed budget and discussed with the customer. This contributes to the Proposal stage. Acceptance stage is where all information, proposals

2. BACKGROUND

and solution documentation with the customer, budget approvals and business acceptance take place for an Opportunity. Negotiation phase is where the customer engages with the sales team raising objections, engaging in next steps etc in order to make progress with the deal. This is also the phase in which the customer has agreed to terms and is finalizing the paperwork to invest in the solution provided. Won stage is where the Opportunity has been successfully completed and contributes to the revenue of the company. Closed/Lost is the stage where the deal fails to materialize. Opportunities are not mandated to go through all these sales stages. It depends on how the deal is progressing and in some stages it can directly go into the Won phase or Closed/Lost phase or can skip some stages too.

2.1.2.2 Forecast Category

As already mentioned in Subsection 2.1.2, Forecast Category is the probability or the likelihood of an Opportunity to complete in an agreed upon time period. The five different categories are shown in figure 2.1. Best Case means an Opportunity has high chances of getting completed. Committed means an Opportunity will be completed in the given time period. Closed means that the Opportunity has already been completed. Omitted refers to Opportunities that gets cancelled, and these deals go out of the sales Pipeline. Pipeline stage is the one at which the Opportunity is present but will not be completed in that agreed time period but will eventually be completed soon. In other words, the deal is delayed but not slipped from the Company's hands.

2.2 Conventional Forecasting Process

The Traditional Forecast process followed in the majority of B2B companies, including the company where this thesis is carried out, revolves around the Sales team. What contributes to the Forecast? As mentioned earlier in Chapter 1, the Opportunities in the Sales Pipeline contribute to the Forecast process. Many companies have different hierarchies as per which they formulate the forecast. Let's consider the procedure followed in the company where this thesis was carried out. Sales forecast is formulated by different personnel at different levels of Sales Territory Hierarchy. The Sales Territory is split into the following ways - District, Region, Multi Region, Area, Multi Area and Geography. For example, the Sales Representatives and District Sales Managers fall into the District level. At each level, with an increase in Sales Territory Hierarchy, there is an increase in the Seniority level of personnel.

The forecast process starts at the District level where the Sales representatives mark the deals which can be included as part of the forecast and submits it to the District manager who formulates the district level of forecast. It moves along the Sales hierarchy up till the Sales Multi Area level where it is finally submitted as the official forecast. At each level, the forecast can be modified based on the person's assumptions and knowledge. Also, the final forecast amount can be adjusted or overridden in order to account for risk or to add any excess opportunities. This finalized forecast which is purely driven based on the knowledge provided by individuals by their interactions with the customer, previous experiences with the customer, knowledge about the existing situation etc is submitted to be the official forecast. The forecast process is carried out every month by the finance team based on the input from the Sales team.

With this background about B2B Sales Forecast process, let's look at the problem and what is our research area, which will set the tone for the remaining part of the thesis.

2.3 Research Problem and Questions

The following points can be observed from the details provided in chapter 2.

- Pipeline is an important part of the Sales process, and it is essential that a company has a healthy pipeline in order to showcase and maintain a strong presence in the market.
- The conventional forecast process is purely driven by Individual's knowledge and assumption about the Opportunity.
- Ability to understand Opportunity conversion rate is minimal, i.e, Will an Opportunity get completed or not?

In theory, the thesis aims to solve an organizational challenge at the company and also aims to provide additional explanation towards the application of Machine Learning in B2B Sales Forecasting process. The following subsections provide the research questions this thesis will answer and the grounds on which it was considered.

2.3.0.1 Research Question - 1

Firstly, We try to solve the organizational challenge faced by the company - NetApp. The company's Financial Planning and Analysis team formulates the sales Forecasts depending on the inputs provided by the Sales team. Although this forecast is of acceptable quality,

2. BACKGROUND

there is a concern that the forecasts are purely driven by the Individual's knowledge. This constitutes to some commonly occurring problems with respect to Sales team and their ability to pick deals to be included as part of the forecast process. Some key problems are:

- If the sales team achieve their sales quota for the month, they can push Opportunities that have high likelihood to get completed to the next forecast period. This means some deals that should be a part of the forecast essentially misses out.
- The ability to mark the deals as part of the forecast process is based on individuals. This can lead to Human bias, repetitive behaviour and other cognitive biases.
- There is no means to challenge the Sales representatives on the deals they selected. It basically becomes a unidirectional process.
- Finally, not all factors are considered when selecting the deals. Limited knowledge is another area of concern, provided the capacity of data available.

This paved the way to look at Machine Learning methods to learn from the Pipeline data and design a predictor which predicts the possibility of the win/loss criteria for each opportunity and also helps in formulating the forecast. This way we can make a data driven approach and also look at an array of features available in the data and pick the ones which are more suitable to solve the problem at hand. Also, new products which are introduced never make it to the forecast in the current year. The forecast value is set to null, as there are no deals in the pipeline for the new product. A survey is carried out to find the expected reach of the product, but it does not get translated into the forecast. This paved the way for the first research question and its sub research questions, which is formulated as :-

- 1. Can a Sales Forecast formulated using Machine Learning challenge the Sales Forecast generated by the Knowledge of the sales team in terms of Quality?
 - Which Machine Learning Algorithm performs better with respect to the prediction?
 - How expert's knowledge can be incorporated into modelling the data?
 - How the model results can be evaluated?
 - What level of accuracy is accepted for the model to be deployed commercially?

2.3.0.2 Research Question - 2

Machine Learning models tend to need constant monitoring to ensure the predictions don't deteriorate over time. In other words, Production models suffer from the concept of Data drift. Data drift is defined as the change in input data compared to that of the training data, which causes the reduced the performance of the model. B2B Sales data has a bigger scope for constantly changing. There are also higher chances for unseen data to flow in as input data for the model. We formulate the following research question to understand the impact of Data drift and also the challenges which arise due to it in making the predictor model production ready.

1. How Data Drift plays a role in B2B Sales Data? Does it have a significant impact, and What can be done to minimize the changes in model performance, if any?

2.3.0.3 Research Question - 3

Building the predictor gives the ability to formulate the forecast procedures, but there is more to Pipeline than just forecasting. Understanding the count of Open Opportunities, count of deals to complete, and various other counts are provided by the tools available in CRM. But the ability to provide advanced analytics that drive the decision-making processes is still a key area for us to research. With the predictor output and the pipeline data, identifying the Key performance indicators (KPI's) and explaining them using Visualizations is needed which leads us to the final research question which is as follows:

- 1. What are the KPI's with respect to Sales Pipeline/Predictor Output and does using Visualizations a good way to understand the Sales Pipeline?
 - What kind of Visualizations can be used for this process?

2. BACKGROUND

3

Related Work

As already stated in the beginning, B2B Sales Forecasting takes up a bigger space in the process of Financial Planning and Analysis, and we try to apply machine learning to perform the process of forecasting. The application of Machine learning in B2C forecasting procedures is higher compared to its usage in formulating B2B forecasting procedures. In general, B2B sales data is quite complicated compared to B2C and the traditional approach of formulating the forecasts as discussed in section 2.2 is favoured more over any alternatives. This chapter gives an overview of experiments and studies carried out to understand and convey the use of machine learning in B2B sales forecasting procedures.

3.1 Machine Learning and B2B Forecasting

B2B Sales forecasting has a limited literature when it comes to the application of machine learning techniques in this domain. This is largely due to the fact that most of the organizations have not yet started exploring the usage of Artificial intelligence in the B2B domain, and also the reluctance to navigate away from the conventional method of forecasting. With that said, we will start looking into the different approaches and problem areas identified by the previous researchers.

Marko Bohanec et al. (2015) proposed a novel model based on machine learning techniques to counter the Sales forecasting problem.(1) They formulate a framework which will involve a classification algorithm at the heart of it and with this try to predict the sales opportunities. The output of the model is aimed to reduce the forecast error and also to provide insights about the model's prediction, which in turn triggers an organizational learning. Thereby, the authors try to create a feedback loop which will involve the machine learning model and via the insights trigger organizational learning. The input data to the

3. RELATED WORK

research was anonymized from the funding company and had only fewer instances of the data. To generate more instances, "semiArtificial" package of R was used to use the data for modelling purposes. RandomForest classifier was observed to be the high performing classifier and recorded accuracies of 96% and the model's output was visualized to form the feedback loop. Although this gave the idea that the machine learning can be applied in B2B sales forecasting domain, yet there were certain limitations. The limitation of this research was it had only limited training instances and much of them were created artificially, which possibly explains the high levels of classification accuracy obtained from this approach. The visualizations used to explain the models are complex which is hard to understand for a person coming from a sales background, which will reduce the participation of people in organizational learning.

Junchi Yan et al. (2015) proposed a model which will build a predictor but will look at things from a regression point of view. The model was aimed to make a predictor which will give a win propensity for each lead. (2) The methodology of this work is based on two steps - First being the collection of historical leads data and also the profile of the leads and the labels associated with the leads. The authors try to explain the synergy between the sales and marketing team and are more focused in determining the propensity score, which will potentially convey whether the lead will be won or not. Stephen Mortensen et al.(2019) decided to create an initial baseline for understanding the sales outcome propensity prediction in B2B setup.(3) They experimented with different classification algorithms like decision tree (4), RandomForest (5) and XGBoost (6) to come up with a good classifier and carried out the implementation using data from a Fortune 500 paper and packaging company. The main purpose of the research in addition to creating a baseline was to find out what features really contributed towards determining the win propensity. Again, RandomForest was found to be performing well with 82% accuracy and precision and recall were around the 77%-79% mark. Also, the authors experimented by providing individual RandomForest models for each business unit of the company and observed that the predictions were better since it operated based on the knowledge within the unit and also the execution time improved.

Brendan Duncan and Charles Elkan (2015) applied machine learning techniques to the B2B sales domain. The goal of the paper is different from predicting the outcome of the sales opportunities. Instead, the authors, try to determine the step before in the sales funnel. They build scoring two scoring models - namely Direct Qualification model and Full, Funnel model (7) which estimates the chance of a lead to get converted into an opportunity. This is more to understand whether the lead generated by the marketing team will be converted into an opportunity, which potentially goes into the won or lost stage to result in the revenue generation process for the company. The research helps to potentially replace the human based lead scoring methodologies used in companies to understand whether a lead will get converted into opportunity or not. Although it is different to the context of the thesis, this still provides a valuable insight into how machine learning is used in the B2B sales forecasting area.

Alirezza Rezazadeh (2020) built an azure based machine learning approach to predict the outcome of sales opportunities. The paper proposes an end to end cloud based workflow to forecast the outcome (8). It takes the binary classification approach and uses a voting based classifier which is built with XGBoost and LightGBM (9) models. The research uses a B2B consulting firm's data belonging to different sectors such as healthcare, energy and finance. The implementation is done using azure and the framework consists of two pipelines - ML and Prediction pipeline which is used for training and for predicting the outcome on the live data. The author tries to formulate the decision boundaries for each of the different sectors and sets them as decision threshold which help to improve the classification accuracy for the individual sectors. The feature generation was another interesting aspect of this research, as the author creates lookup tables containing statistics for the categorical variables and merges them back to the original dataset at hand to create new features. The evaluation of the model's prediction was also slightly different. In addition to the classification metrics, the author makes use of the user generated probability score for each opportunity and compares them to the probabilities generated by the model to evaluate the outcome.

A forecast framework consisting of different types of machine learning models that would work based on different sales patterns or data changes was proposed by Xin Xu et al. in 2017. At the heart of the forecast engine lies different models namely - Probability models, Opportunity score aggregation models, time series models, Hybrid models and Neural network models and the optimal model is selected based on the request generated by the user(10). The probability models work to provide an aggregate level of forecast by grouping opportunities into time periods or groups preferred by the user, whereas the opportunity score aggregation models look into the history of the opportunity and calculate the likelihood that it can be closed within the time period specified. The time series also works on the aggregated level with respect to date/time/year. This methodology produced higher forecast accuracy levels, but the forecast and predictions happen at an aggregate level and not at an opportunity level. Timo Thiess et al. (2020) designed a predictor to find out about the B2B sales outcome of the opportunities as part of a project with MAN Energy Solutions (11). The research focuses on building a predictor model, but intends

3. RELATED WORK

to extend the explainability of the machine learning techniques in the B2B sales area. LightGBM classifier is used to classify the outcome of the opportunities and SHAP (12) is used to explain the machine learning model. The author also formulates certain steps which are termed as the design principles upon which their framework was built. The author with the help of SHAP and an interface explains the model's predictions for each opportunity with the intention to improve organizational learning.

With the brief look into the literature, we were able to find some common limitations which we try to overcome as part of our thesis in addition to solving the organizational challenge at hand. Research in this area often used limited training data or semi artificial training data, which eradicated many real life transformations happening in the nature of the Sales pipeline data at an organization. This might also influence the accuracy related to the classifications. Next, we were able to find that the some researches mentioned above, formulated the forecast value, but they didn't talk about the quality of the data except for one. For example, If the quality of the data is low on the sales value of the opportunities, then it will lead to a classifier that is performing exceptional at classifying the opportunity's outcome but not so much when it comes to getting the forecast value. Additionally, the feedback learning is a good approach to improve the model's performance and the thinking of the teams in context to improving the way they look at the outcomes, but it doesn't shed any light on the status of the sales pipeline and the quality of the data in the sales pipeline which can also be improved with the help of visualizations. So with all this and even some smaller limitations observed in mind, we formulate our modelling approach in the following chapters.

Data Overview

4

This chapter deals with the details about the B2B Sales dataset which was used. We try to initially provide an explanation about the data sources and then more into how it was transformed from its raw format into a processed input which could be fed to the machine learning pipeline to generate predictions. In this chapter, we also provide insights into how feature engineering and encoding methodologies were used and whether they yielded any results or not.

4.1 Data Explanation

B2B Sales dataset consists of data related to the Sales Pipeline. Our B2B Sales data is provided to us by NetApp. The Sales representatives enter quote details about prospective deals into the Quote interface system developed by a third party. So the dataset basically consists of upcoming, completed, missed and lost opportunities along with opportunity specific details. The data is stored in a CRM system which houses all levels of details associated to the data - Personal, Sale details, Product details, Customer details etc. We pull the data from the CRM and store it into a relational database and query this table to get the training data. This arrangement of connecting them to a relational table also has a reason behind it. For the machine learning aspect of the thesis, which will be explained in the later sections, We need to have a training pipeline and a production pipeline that influences us to make this decision of pulling the data from CRM into a relational database. As mentioned in various (reference papers), the sales data is noisy. The main driving factor behind this is that the values are updated by the sales people, which brings in the possibility of introducing human errors like abnormal sales value, incorrect updates to critical fields and ability to leave fields unfilled. The B2B Sales data collection is a purely human driven

4. DATA OVERVIEW

approach, and it's not something like sensor data where the devices collect the data and the margin of error is lower.

The raw dataset obtained from the CRM consists of 179 features pertaining to different levels of information related to an opportunity. Although there were many dimensions to the dataset, We didn't make use of most of them as they did not provide any valuable insight to our goal. Upon discussion with the responsible people, we trimmed down the dataset to contain only 34 features, including both dimensions and measures. The reasoning behind the trimming down of the dataset is provided below:

- Most of the features were duplicate features, and they were marked redundant. This was due to the pending clean up of the CRM data.
- Many features reflected Opportunity value in different currencies, as the dataset contained worldwide data. As the forecasts were formulated in US Dollars, only values in USD was considered and the rest were removed.
- Drill down into product level details, areas, customer segmentation types were also present which were excluded.

Description	Value
Dataset Size	0.3M
Total Features	179
Selected Features	34

Table 4.1: Dataset - Size/Shape

4.2 Train and Test Data

Every machine learning algorithm needs data to produce results. General practice is to split the dataset into Training and test datasets to train and evaluate the model's performance. In our scenario, we have the historical pipeline data from the First quarter of 2018 to the third quarter of 2021 which was used as the training data and the fourth quarter of 2021 was used as the test dataset.

Description	Value	Duration of Data
Training Dataset	215.6K	2018Q1-2021Q3
Test Dataset	92.4K	2021Q4

Table 4.2: Train/Test - Details

4.2.1 Train Data Profiling

The training data is profiled to understand its composition. The profiling is explained in short below:

- Overview :- The training data consists of 215.6K records as stated in the previous section with 34 features. The dataset possesses a label which conveys whether an opportunity was won or lost. We have zero duplicate records and the missing cells % i.e, the number of cells found to have no data in the overall dataset seems to be 5.5 %. The dataset takes up a memory size of 94.4 MiB in total.
- Categorical Features :- Sales datasets always have higher number of categorical features as organizations use them to house the Sales territory details such as country, region etc and also details such as product type, Customer type too. In the training dataset, we have 29 Categorical features - Seven of them are highly cardinal.
- Figure 4.1 below shows the amount of missing data present in the dataset. The figure shows the count of values present in the dataset, split by different features. Figure 4.2 reflects the heatmap displaying the correlation between the variables.

4.3 Feature Engineering

This section explains in detail the preprocessing steps, encoding methodologies and the outlook of the processed input data.

4.3.1 Data Preprocessing

In the Data Preprocessing stage, the aim is to prepare the raw data into the desired format and clean the dataset. The following steps were carried out as part of the preprocessing stage:

• Firstly, Unknown values in features representing Sales territory and Sales Team details were removed to.

4. DATA OVERVIEW



Figure 4.1: Missing Data - Split by Features

• With human based input, We can expect mistakes in Product names. So as part of this step, mapping is done to ensure correct product names and product classifications

are reflected all across the dataset.

• Since the pipeline is used for also testing purposes, It contains some test opportunities. These are marked out as test under description and they are filtered out from the dataset as well.

4.3.2 Adding New Features

As part of feature engineering, we create new features from already existing features with the assumption that they will help produce better output. Features created reflect business knowledge and the cyclical aspects of an opportunity. Table 4.3 reflects the new features added and their description. The target feature "Label" is also created in this step which conveys whether a deal was won or lost, which is the expected outcome.

As a final step we keep the newly created features and drop unwanted features that convey the same information, i.e., duplicate features.





Figure 4.2: Heatmap - Displaying correlation

Feature	Description	
Duration	Time taken to create and close an opportunity (in days).	
Install Base	Classification between New Customer, Technology Refresh or renewal based customers.	
Distributor/Reseller/Sold to Partner	Separate features explaining the medium of the sale	
Quarter/Month/Week	Date features derived from the closing date of an opportunity.	

 Table 4.3: New Features Description - Details

4. DATA OVERVIEW

4.3.3 Categorical Encoding

Categorical Encoding is highly important to improve the model's performance, and also the ability to translate the same level of knowledge from categories to encoded values is important. This essentially means that the model needs to be able to interpret the meaning of the categorical features correctly, which leads us to effectively use the categorical data we have at hand. When it comes to categorical data, there is no one possible pathway to success. It is a trial and error method and one needs to experiment with different encoding techniques to see which improves the performance of the specific machine learning model to use. There are some commonly used Categorical Encoding techniques such as One Hot encoding, Nominal encoding and Ordinal encoding.(13) In particular, One hot encoding is widely used when the categorical features doesn't convey any information about order and doesn't need to be represented in an orderly manner. As mentioned in section 4.2.1, the training dataset consists of High cardinality features as well. The problem with High cardinality features is that they come with some difficulties owing to the different values present in them. It's difficult to apply the common techniques like one hot encoding or label encoding as they lead to introduction of sparsity in the dataset or lead to high number of unique values which fails to convey the meaning of these values. (14) To combat encoding high cardinal features, there are some advanced encoding methodologies such as Weight of Evidence encoding, Target Encoding, Mean encoding etc. Also, we have some Machine learning algorithms which can automatically encode the categorical values as per their own encoding scheme, such as the XGBoost encoder and CatBoost encoders.

With all this in mind, we apply different encoding schemes to the categorical variables and compare their performance on different classification algorithms. Upon doing this, the results show that applying CatBoost encoder provided better performance in terms of accuracy and f1 score for our dataset. Table 4.4 shows the performance metrics after the application of the CatBoost encoding technique on different classification algorithms.

ML Algorithm	F1	Accuracy
RandomForest Classifier	0.77	0.77
XGB Classifier	0.76	0.77
CatBoost Classifier	0.77	0.77
DecisionTree Classifier	0.71	0.72

 Table 4.4: CatBoost Encoding Performance
4.3.4 CatBoost Encoding

CatBoost encoding is a target based categorical encoder. It is similar to the target encoding, but avoids the problem of target leakage by means of an ordering principle. CatBoost in general computes target statistics based on the ordering principle. The method takes a random permutation of the training samples and for each sample it computes the target statistics by using all the available history. This essentially means the target statistic for a categorical feature is calculated only from the observations before it, which helps to overcome the problem of target leakage. (15) (16)

CatBoost encoding takes into account prior value, which is common practice and helps to reduce noise from low frequency categories. For binary classification methods, the prior is a priori probability of encountering a positive class. This calculation methodology makes it to be flexible when encountering new features as well.(17)

$$\frac{(TargetSum + prior)}{(FeatureCount + 1)}$$

where TargetSum reflect the sum of the target for that particular feature value, prior is a constant calculated at the start and FeatureCount is the total of categorical features observed with the same value as the current one.To understand the encoding methodology even clearer, we introduce an example. Suppose we have a categorical feature "car" with values like ['BMW','AUDI','AUDI','BENZ'] and target column as "Bought" with values ['1','0','1','1']. From this as an initial step, we calculate the prior to be 3/4 = 0.75. For the category value 'AUDI', the target count will be 1+0 = 1 and the feature count will be 2 as there are two instances of that value. So the encoded value for 'AUDI' will be (1+0.75)/(2+1) = 0.58. This is how the encoding is calculated in general if we consider this as one permutation.

The CatBoost encoding is implemented using the CatBoost library along with the Cat-Boost classifier. There are several other implementations, with the most common one being using the category encoders library of python, which provides implementations for a wide range of encoders.

After carrying out all the processes as mentioned in the previous sections, The data is now ready to be used for building the model. The table 4.5 shows the final list of features that were present before applying feature selection techniques on them.

4. DATA OVERVIEW

Group	Features
Date Based	Quarter, Month, Week, Duration
Product Based	low/high flash, low/high PCS, low/high Hybrid Cloud, PS , Renewals, Others
Sales Territory	Sales Area, Sales Multi Area
Customer	Customer market segment, Wallet Range, Install Base
Opportunity Detail	Distributor, Deal size, Sales channel, Created by user type, Sales Team, Amount

Table 4.5: Final list of Features

4.4 Feature Selection Approaches

Feature Selection is an important step in machine learning. It helps to identify important features and also helps to disregard unimportant features. This contributes to improvement in model's performance. The literature on feature selection methodologies states three different types of feature selection approaches(18)(19). The three common approaches are

- Filter Methods: The filter methods are based on statistics and use an approach which is completely independent from using a machine learning algorithm. The filter method uses rank ordering method for variable selection. Techniques such as Chi-square test, Pearson correlation criteria, Mutual information etc. are part of the filter methods. Since they involve statistics and are based on ranking methodology, the filter methods tend to be computationally faster even with high dimensional datasets due to their tendency to scale and also avoids the problem of over fitting. The main disadvantage is that it considers each feature individually and scores them which can lead to degradation of the classification performance as it doesn't take into account the interaction of variables.
- Wrapper Methods: The wrapper methods are another type of feature selection techniques. The wrapper methods are dependent on the machine learning algorithm to select the features. They select the features that help to improve the classifier's performance in terms of classification accuracy and this makes them classifier dependent methods. Recursive Feature Elimination (RFE), Backward Feature Elimination (BFE) and Boruta (20) are some of the most common wrapper methods. The wrapper methods are computationally expensive when compared to filter methods. They become more expensive when the size of the dataset increases. Although the wrapper

method takes into consideration the interaction between variables, it can still cause over fitting as it strives to improve the classifier's performance.

• Embedded Methods: Embedded methods are a combination of wrapper and filter methods. They combine the qualities of both and are implemented by all machine learning algorithms as an in build feature selection approach. RandomForest, Cat-Boost, Decision Tree, Linear SVM are some of the algorithms that have their own inbuilt feature selection capabilities. The benefits of the embedded methods are the models results are interpretable, and they generalize better.

As mentioned, all of these different methods have different advantages and disadvantages. With the help of the correlation heatmap shown in figure 4.2 and also as mentioned in subsection 4.3.1, we remove certain features that were highly correlated and also those that were conveying the same information. As part of the thesis, Wrapper methods like Recursive Feature Elimination and Boruta Algorithms were tried out, but they were not chosen due to their time complexity. With a larger training data, we experienced the feature selection task using wrapper methods to be time-consuming and computationally expensive.

For the thesis, The approach was to use a combination of Filter and Embedded methods and choose the feature with the highest votes, i.e, gets picked by the different algorithms as important. The methods used are listed as follows

- Chi Square Test It is a statistical test applied to the groups of categorical features to evaluate the likelihood of correlation or association between them using their frequency distribution.
- Random Forest RandomForest classifier is used for this method. Random Forest uses its inbuilt feature selection to evaluate the model, and it works on the principle of how pure each of the buckets containing similar data but different from one another. For classification problems, the model uses Gini impurity or Information gain/entropy.
- CatBoost Classifier CatBoost Classifier also is an embedded method with its own inbuilt feature selection. It works on getting the feature importance by means of understanding the weight of the leaves in the model.

Table 4.6 shows the feature scores along with the different feature selection algorithms. "1" conveys it's an important feature, and "0" means it can be removed. The "final score" column shows us the overall scores for each feature. The idea initially was to take the features which scored greater than or equal to two and apply them to the classification algorithm. Results from that approach showed that removing all features with one as the score resulted in a volatile model performance. So after keeping and different features, we decided to remove the product grouping features as those were the ones producing the changes in performance. The features is_PS, is_Renewals, is_High_Flash, is_low_flash,is_high_PCS, is_low_PCS and is_others were removed. Features such as Deal size, Install_Base, Distributor, Created_by_user_type and End customer market segment were kept even though they had a score of one.

Features	\mathbf{RF}	Catboost	Chi	Score
Duration	1	1	1	3
Wallet_range	1	1	1	3
Sales_Team	1	1	0	3
Sales_Area	1	1	0	3
Month	1	1	0	3
is_PS	0	0	1	1
is_others	0	0	1	1
is_Renewals	0	0	1	1
Sales_channel	1	1	0	2
Sales_multi_area	1	1	0	2
Sales_channel	1	1	0	2
Deal size	0	0	1	1
install_base	0	0	1	1
is_high_flash	0	0	0	0
is_hybrid_cloud	0	0	0	0
is_low_flash	0	0	0	0
is_low_pcs	0	0	0	0
endcustomermarketsegment	0	0	1	1
distributor	0	0	1	1
created_by_user_type	0	0	1	1
is_high_pcs	0	0	0	0

 Table 4.6:
 Feature selection Approaches

4. DATA OVERVIEW

$\mathbf{5}$

The Hybrid Approach - Machine Learning with Simple Maths

This chapter explains the approach taken to solve the problem at hand, the design of the model where we try to distinguish between how the training and the active pipeline works and further details about the workflow of the model.

5.1 Overview

There is always an end goal which is to be reached with every research. For this work, The end goal is to predict whether the opportunities will be won or lost and to formulate an expected forecast value at the Sales Multi Area level. With Sales Multi Area level, the expectation is to get the forecast values at an EMEA, Americas and Asia Pacific level. Additionally, we provide visualizations to explore the historical data and the actual progress of the Sales. To facilitate this, the design consists of three components.

- Machine Learning : At the heart of the design is the Classification Machine learning algorithm Predictor. There are different ways to approach this prediction problem. It can be considered as a Classification or as a Regression problem. As mentioned in Chapter 3, There have been related work done considering the problem to be either of the above two. With that said, We take the Classification approach instead of the regression approach. We treat the target as labels, with values as Won or Lost. This provides the answer to the first part of our goal, which is to predict whether an opportunity will be won or lost.
- Mathematical Approach : The next stage is to formulate an accurate forecast value at the Sales Multi Area level. For this, we consider only the won deals from the output of

the machine learning model. The problem is that the sale value for some opportunities are not accurate. There are chances for opportunities to have zero or very high values in the sales pipeline. Although we exclude opportunities with abnormally high values in the preprocessing stage, we cannot do the same for opportunities with zero values. So when we aggregate the sales value at a Sales Multi Area level, it leads to higher or lower forecast value, which in turn triggers the major problem of Over forecasting or Under forecasting. This will also be explained in detail in the later sections of this chapter. To overcome this, we employ a simple mathematical approach which takes in as input the output data from the machine learning model and the Historical Sales data from the past, which only contains deals that got converted into revenue for the company. With the help of a weighted average, we calculate the forecast values for each multi area, with which we find a solution for our second part of the end goal.

• Visualizations: The final component is the Visualization. The historical sales pipeline needs to be monitored to check its hygiene and health aspects. Also, we need a medium to communicate the output of the machine learning model. The solution to the third part of the end goal lies with Tableau. We use tableau to create dashboards that convey the information required to monitor and assess the sales pipeline.



Figure 5.1: Design Components

5.2 Machine Learning Algorithm - Classification Model

Our approach falls in the line of Binary Classification, where we classify the opportunities into either won or lost stage. Binary Classification is the task of classifying elements into one of the two available groups (21). Various Binary Classification algorithms have come into picture ever since the increase in the usage of machine learning to solve real life problems. Some ideal examples of binary classification problems are marking an email as spam or not, identifying whether a person has a disease or not etc. Different types of classifiers are in practice these days such as linear based classifier, Tree based classifiers, Support vector machines and much more complex classifiers based on neural networks. Ensemble methods (22) and Gradient boosting algorithms are also a welcome addition to the list of classifiers. Some of these classifiers also extend support to Multiclass classification, where the elements are classified into more than two groups. An ideal example for the multiclass classification would be the prediction of weather type such as sunny, windy or rainy. The underlying concept behind binary classification (21) is that the elements are applied to a classifier and the classifier calculates a prediction score which potentially reflects the ability of the element to be classified as the positive class. The algorithm after predicting the score compares it to a classification threshold and if the predicted score is greater than the classification threshold then the element belongs to the positive class and vice versa. For our thesis, we take a closer look at Gradient boosting algorithms, as they were the ones which performed superior in terms of classifying opportunities as won or lost.

Gradient boosting algorithms train a sequence of weak models, where each of them compensate for their predecessor's weakness.(23) The idea behind gradient boosting is to choose a weak model which can be a decision tree or a regression model and build on top of it. Most popular types of Gradient boosting algorithms are AdaBoost, XGBoost, LightGBM and CatBoost. XGBoost, LightGBM and CatBoost are the most widely used boosting algorithms as they give a greater model performance and are faster compared to the other algorithms present. For the prediction purposes, we used the CatBoost Classifier - a supervised classification algorithm based on Gradient boosting. The motivation behind the idea to use CatBoost instead of other algorithms was because of the following reasons.

- Firstly, The model gave a superior classification accuracy, and it produced higher scores in terms of f1-score when compared to the other models as shown in table 4.4.
- CatBoost also offers support to categorical features. Since our data is highly categorical, this ensured good results. As discussed in the Challenges section 7.1, There were also some problems faced with respect to categorical data that were handled well by CatBoost.

5.2.1 CatBoost

CatBoost is a machine learning algorithm that uses gradient boosting on decision trees. It was developed as an open source machine learning algorithm library by Yandex in 2017.

5. THE HYBRID APPROACH - MACHINE LEARNING WITH SIMPLE MATHS

The CatBoost Classifier is one of the implementations of the CatBoost algorithm, which aims to solve any Binary Classification and Multiclass Classification problems. The Cat-Boost algorithm is stated to be performing highly with respect to other gradient boosting algorithms like XGBoost, LightGBM etc. There have been benchmark tests performed with CatBoost and other gradient boosting algorithms which have conveyed the same. This makes us to understand what makes CatBoost better when compared to other algorithms.

Literature based on CatBoost from the authors of the algorithm explains in detail how CatBoost works (16)(15). CatBoost as the name states is the acronym for Categorical Boosting and the algorithm specifically performs better in terms of handling and encoding categorical features with different methodologies when compared to others. The algorithm makes use of permutation techniques, One hot max size and target based statistics to handle the categorical features. The CatBoost algorithm divides the dataset into different subsets and encodes the categorical values using one hot encoding, target statistic or count based depending upon the specifications of the model. The algorithm avoids overfitting and target leakage by the concept of ordered boosting, for which a hint of what it does is provided in the section 4.3.4. The other advantages of CatBoost is the ability to produce great quality without parameter tuning, along with faster training and prediction time. This was also critical for us as we worked with limited resources for the thesis. The thesis implements a default CatBoost classifier with the learning rate specified at 0.1 with all other features set to their default values. This also highlights the aspect of the algorithm where it can produce high quality without parameter tuning, as it performed better than all other classifiers on the B2B sales data.

5.3 Mathematical Approach

The mathematical approach is taken to get an accurate forecast value at the Sales Multi Area level, which is one of the goals of the thesis. A key motivation to add this additional layer on top of the machine learning model was discussed in brief in the section 5.1. As mentioned in section 5.1, the sales value associated with an opportunity determines the accuracy of the forecast that can be formulated. Listed below are the key factors to pick up this approach:

• The sales value can be zero in the sales pipeline data, but can have a significant value in the actual sales data. For example, an opportunity "XYZ" of a customer "ABC" can have 0 as its booking amount in the sales pipeline. This opportunity is a won deal. So there is an entry which is made in the actual sales dataset which houses only deals that are won by the company and this is absolute truth in terms of calculating the sales at the company. The same opportunity "XYZ" has a value of 200,000 euros in the actual sales dataset. The predictor uses the sales pipeline data and even if the predictor identifies this deal as a won deal, the forecast formulated at the end cannot account for the difference of 200,000 euros which leads to an under forecast. Similarly, the opposite of this use case is also possible that can lead to over forecast.

• Usually, the classification accuracy of a model can be improved with the help of a confusion matrix. By varying the threshold level, we can reduce or increase the True positive or True negative. This is usually the case when we try to apply machine learning in areas such as spam filtering or cancer detection, where we need to be account for the maximum of true positive cases and there can be occasional misses. But the same concept cannot be applied in the B2B sales forecasting area, as this again creates the problem of over/under forecasting. If we account for either one, it leads to a biased output.

To overcome these, we employ the mathematical approach. The input to this is the output data of the machine learning model as stated in section 5.2, the actual sales data of the company over the years post 2020. Figure 5.2 shows the steps on how the mathematical approach is performed. The approach places weight on the historical actual sales data. It considers the previous year and gives it a higher weight to account for recent trend, and it also considers the value from two years back and places a weight on it to account for consistency. Figure 5.3 explains the steps with an example.

5.4 Visualizations

The visualizations constitute the final component in the design of this model. Tableau, the popular business intelligence tool, is used to create the visualizations. The visualizations are aimed at providing insights into the sales pipeline's Hygiene and Health. Sales pipeline keeps constantly changing with respect to time. There is always opportunities flowing into and out of the pipeline. Since this is used to formulate the forecasts upon which the companies build on their plans for the upcoming periods, it's essential to monitor the sales health and hygiene aspects closely.

5. THE HYBRID APPROACH - MACHINE LEARNING WITH SIMPLE MATHS



Figure 5.2: Mathematical Approach to formulate forecast - Steps

Hygiene of the Sales pipeline refers to the process of keeping the data as accurate and up to date as possible. This refers to the process of checking the opportunities for the following:



Figure 5.3: Mathematical Approach to formulate forecast - Example

- Should reflect correct close date.
- Should be tagged with the correct products present in the product portfolio.

- Should reflect reasonable sales value for opportunities.
- Ensuring the opportunity details get updated at each stage or upon any change in status of the opportunity.

Maintaining a good hygiene will result in improved forecast accuracy, helps lay the foundation for analytics, and also helps sales representatives learn about the opportunities. Although it has advantages, Hygiene is one of the hardest things to track, as there is a common misconception that spending a lot of time on entering and updating the records leads to less time for the sales professionals to carry out sales activities. Also, there is an impression that it doesn't help the sales teams, which leads to failure in maintaining good hygiene levels.

Similarly, Health of the Sales pipeline refers to the assessment of the pipeline to identify warning signs or achievement's in one's business. The health of the sales pipeline is monitored to ensure the following:

- To ensure that there are enough opportunities in the pipeline to reach the estimated value in the annual operating plan.
- To understand the success rate at particular points of time Win/Loss monitoring of opportunities,
- Duration taken to close opportunities on an average.
- Identifying opportunities that slipped, i.e, those deals that were to be completed in a promised time period but didn't materialize and instead they were shifted to a newer time period.
- Understanding the reasons behind why an opportunity is lost.

Hygiene and Health are related. If there is a good hygiene for the sales pipeline, then it will enable to drive good health analytics on a daily/weekly/monthly basis which can help drive discussions and help improve the organization's business capabilities. Along with this, we also need a visualization to communicate the output of the machine learning model and the output of the mathematical approach. Tableau is used to develop the dashboards, which will help provide insights into the three different tasks at hand. We design four different dashboards that help to solve the above problems.

5.4.0.1 Predictor Output Dashboard

This dashboard is composed of two views. The first view shows the output of the mathematical approach. As mentioned in section 5.3, the mathematical approach helps to formulate the forecast value. So this view displays the forecast value aggregated and adjusted at a sales multi area level. The second view lists the opportunities and provides the predicted output associated to each of them. Each opportunity is displayed with its ID, Name, Multi Area, Area, Customer name, Product name, booking amount and its predicted label. Both of them are plotted in the form of tables. It only reflects the details for the selected forecast month.

5.4.0.2 Pipeline Hygiene - Active

This dashboard is used to display the details regarding the hygiene of the active pipeline. It consists of four views. The first view is a circle view that identifies Opportunities above a certain threshold sales value, which helps in filtering the opportunities with unreasonable sales values. The second view is a bar chart with each bar representing the count of opportunities with zero as their booking amount and are grouped by the products present in the product portfolio. The third view is again a bar chart which shows the amount of missing values present in critical columns which are key to monitor the sales pipeline. The final view is a tree map which shows opportunities with a close date lesser than their creation date.

5.4.0.3 Pipeline Health - Active / Past

Two dashboards are used to represent the pipeline health for the Active sales pipeline as well as the past sales pipeline.

- Active Sales Pipeline Dashboard: For the current month, This dashboard explains the deals in the pipeline, which gives a sense of understanding about the business for the month. The dashboard churns out information such as the count of opportunities won and lost in the period which are looked at from different dimensions, lists the opportunities by different stages of the opportunity lifecycle as shown in figure 2.1 and finally shows the sales for that month over the years to give an idea of how it's doing year over year.
- Past sales Pipeline Dashboard: With emphasis to reflect on the performance over the past years, this dashboard helps to understand how functional the sales teams were

and how the sales took place. So this primarily shows the performance of Sales teams using the number of opportunities they closed, lost and their conversion percentages. This also helps to take a look into what path the sale took, whether it was directly sold or whether the sale happened via different resellers and distributors. The final part of the dashboard shows the pattern of won/close deals over the years for each month, with which one can get a clear understanding of what kind of sales pattern to expect.

Thus, we find the solutions for the three part goal which was explained in the section 5.1. In the following section, we explain the two different pipelines used to achieve the experimental portion of the explanations in the previous sections.

5.5 Model Flow

The architecture of the model as shown in figure 5.4 takes the approach of having two different flows - the Training flow and the Active flow. The following subsections talk in detail about the different pipelines and how we set them up.



Figure 5.4: Predictor Architecture

5.5.1 Training Flow

The training flow is made up of all the different components mentioned in Chapter 4. The historical data is queried from the database, which is then split into train and test data for applying machine learning techniques. The training flow consists of two branches as shown in figure 5.4 depending on the data being used. Branch one uses the train data and performs the preprocessing as mentioned in section 4.3.1 and feature creation as mentioned in section 4.3.2. Post this feature selection is applied as mentioned in section 4.4 which helps us to choose the best features to use for the predictions, and we also stated that we take some features which are not considered important by the feature selection techniques. The filtered data is then fed to the different classifiers from which CatBoost was selected. Similarly, the feature creation and preprocessing steps are carried out on the test data and this is directly passed to the CatBoost classifier. This pipeline is created, but there is no scheduled run to retrain the models. Evaluation of metrics is taking place constantly, and the process of retraining is triggered manually only when there is a drop in the performance or if there are some changes on the data side. The implementation is completely done using Python and the data querying was done using SQL. The libraries of Python that were majorly used to bring this flow to working mode are listed in the table 5.1.

Function	Libraries		
Data Loading	pyodbc		
Data Preprocessing	Pandas, numpy		
Feature Selection	sklearn		
Classification Algorithm	sklearn, catboost, xgboost, lightgbm		
Monitoring	Evidently.ai		

Table 5.1: Python Libraries

5.5.2 Active Flow

The production based model which is used by the team makes use of the active flow. The active flow uses the new data flowing into the system. All the components in the active flow are packaged into a flask application which is run on a production server. An interface is present in the front and when the user clicks on "Generate forecast", it triggers the

5. THE HYBRID APPROACH - MACHINE LEARNING WITH SIMPLE MATHS

rest of the processes in the background. The data flowing into the system is stored in a table in the database and the table is truncated and created every month to reflect the Sales pipeline data for the current month which is to be forecasted. A scheduled SQL job keeps this table updated with the new data. This data is then preprocessed, and the new features are created as well, and this data is fed to the CatBoost classifier selected from the training flow. This predicts the outcome of the opportunities and this in turn triggers the mathematical approach which calculates the forecast value at a Sales multi area level. Finally, the output data of the predictor as well as the mathematical approach is written back to the database. This output table is used in tandem with the historical sales pipeline, historical sales data and the current month's pipeline data to create the dashboards in Tableau. The classification model is periodically evaluated every month by an automatic performance profiler to identify any deterioration in the model performance. Also, this helps to make a decision on whether the model needs to be retrained or not.

This whole process takes around 2-3 minutes for each click on the "Generate forecast" button as shown in figure 5.5. The whole flow is implemented using python and the whole web application is containerized using docker. The web interface is built using HyperText Markup Language (HTML) and Cascading Style Sheets (CSS). The container created is deployed into production using Kubernetes and is monitored constantly to support in the event of a crash.



Figure 5.5: Infer - interface

6

Experiment & Evaluation of Results

The experiment carried out is discussed in detail in this chapter. Evaluation of Results is a key component that helps us to understand the outcome of the experiment and the trust level behind the results. This chapter talks about the classification metrics we looked at and the accuracy behind the forecast results. Also, it takes a look into how effective the dashboards were in terms of discussions among team members and the action points they could derive from them.

6.1 Experiment

From the training flow, The explanation of the training data is already provided in section 4.2. We perform a K-Fold cross validation (24) with K as 10 to ensure whether the model is consistent in terms of performance and also to avoid the problem of building a model that is overfitting/under fitting. There is a high volume of data and this drove the motivation to use a K-Fold cross validation to evaluate the model performance. The commonly used approach which is the ratio based train test split method can be used here, but it always has the problem of what kind of data it takes for the test split. Although, the split can be controlled in terms of balancing the data samples taken by the method, yet we can't really say that the model is generalizing to all samples of data. Maybe there is chance that the model performs well for the divided data or maybe not and this makes us to use K-Fold cross validation which helps minimize this uncertainty. Post this split, the execution of the training flow happens, where we prepare the data and fit different classification algorithms and evaluate the performance of the active flow. The entire process is implemented as a pipeline using the sklearn library. However, we cannot make use of the

same functionality for CatBoost due to support issues. So we handle the cross validation using CatBoost individually with the model's inbuilt capabilities.(25) The training flow also consists of a test data. With K- fold cross validation, we get to know the overall model performance on the training dataset and this only gives us an idea of how the classifier performs on the dataset in terms of predicting the opportunities as won or lost. Although it ticks the evaluation criteria for one of the thesis goals, we also need to understand the forecasting accuracy, which is the main expectation. So, as shown in table 4.2, we have a separate dataset which contains data only for the fourth quarter of financial year 2021.We make use of this dataset as it makes way to compare the obtained forecast values and the predictions with the actual sales values for this period. At the time of doing the thesis, the fourth quarter has already passed, So we took all closed and won deals from this quarter and used them as the test set. Although the test set is formed differently compared to usual approaches where we just use the Train Test split methodology, this way of formulation was required to obtain a clear picture of how the process was performing.

6.2 Evaluating CatBoost Model Performance - Chosen Performance Metrics

As mentioned in section 5.2, Different classification algorithms were applied to the get the right classifier for our data. Since we employed the K-fold cross validation method, we average the different estimates of the classification metrics for each classifier and use that as the scoring metric to finally select the best model. Table 4.4 already states the scores of the different classifiers, and the section 5.2 discusses why we chose CatBoost as the best fit for our use case. This section evaluates the performance on the test data from the training flow, which is described in the section 6.1. We formulate the performance metrics based on the three different components shown in figure 5.1 to evaluate the model performance.

6.2.1 Classification Metrics

Evidently AI - a python library (26), helps to evaluate the model performance on both the active flow and the training flows (section 5.5). The library provides different automatic and inbuilt reporting techniques to help monitor the machine learning models in production as well as in development. This can be only used on tabular data and not any other forms of data. We also make use of the CatBoost's inbuilt evaluation tools to obtain some additional information about the performance of the classification model. The evaluation happens automatically as part of the active flow, but it is trig erred manually in terms of the

6.2 Evaluating CatBoost Model Performance - Chosen Performance Metrics

training flow. We make use of the classification report generated by Evidently and take the data points from them to evaluate the model. The report is based on the performance of the model on the test dataset of the training pipeline shown in figure 5.4. The class representation of the test dataset is shown in figure 6.1 which shows a fairly balanced dataset. The amount of lost opportunities is a bit higher than the won opportunities, which aligns perfectly with the business understanding. From that report, we look at the



Figure 6.1: Class representation of Test Data - Training Flow

following metrics listed below. As mentioned in the previous sections of the paper, the aim of this classifier is to predict both the won opportunities and lost opportunities efficiently. There cannot be a scenario which can be accepted in which the model performs extremely well for True positives and doesn't perform well in identifying the true negatives. This is because the ultimate aim is to predict the number of opportunities which will be won or lost, and business needs to have a proper understanding of both in order to work towards improving in the future. Accounting for either True positives or True negatives will lead to the problem of under/over forecasting and if that happens, it will lead to the downfall of the entire idea behind the project. Keeping that in mind, we identified the following metrics to evaluate the classifier performance.

• Confusion Matrix: Confusion matrices (27) are the most commonly used evaluation metrics when it comes to determining a classifier's performance. It gives a summary of the predicted results of a classification by giving the listing out the number of correct and incorrect classifications the model made. The CatBoost classifier's confusion

matrix on the test dataset of the training pipeline is shown in the figure 6.2. From the confusion matrix, it's quite clear that the model is making a good generalization when it comes to predicting the correctly won(29091) and lost deals(42844). Also, from the confusion matrix, we can observe the opportunities that were misclassified. The conclusion is that the model minimizes the impact of false positives when compared to the impact of false negatives, but still there is a significant amount of opportunities getting misclassified.



Figure 6.2: Confusion Matrix of Test Data - Training Flow

• Precision, Recall, F1-Score and Classification Accuracy: The classification accuracy measures the ratio of correct predictions over the total number of instances. The precision and recall measures are also taken into consideration, where precision gives the ratio between the true positives and all the positives present in the data and recall is the measure of us identifying correctly the true positives.F1-Score is the harmonic mean of the precision and recall. (28) The equations 6.1, 6.2, 6.3, 6.4 represent the way in which the terminologies are calculated.

$$Accuracy = (TP + TN)/(TP + FP + TN + FN)$$
(6.1)

$$Precision(P) = (TP)/(TP + FP)$$
(6.2)

$$Recall(R) = (TP)/(TP + FN)$$
(6.3)

$$F1 - Score(F1) = 2 * ((P * R)/(P + R))$$
(6.4)

The values recorded for the classifier model are shown in table 6.1. To account for both the precision and recall, we consider the F1 score, since our goal is to have a good precision as well as recall. The model performance is overall at the 77% mark for all the performance metrics discussed, which is acceptable in terms of classifying the win/lost opportunities. Figure 6.3 shows the quality metrics by Class for the test dataset - Metrics by label value.



Figure 6.3: Quality Metrics by Class for Test Data - Training Flow

Metric	Score
Accuracy	0.778
Precision	0.775
Recall	0.772
F1 score	0.773
AUC Score	0.772

 Table 6.1: CatBoost - Classification Performance Metrics

6. EXPERIMENT & EVALUATION OF RESULTS

• AUC-ROC: AUC-ROC Curve (27) is a performance metric which gives an overall ability of the classifier to classify as 0's as 0 and 1's as 1. Unlike the other metrics mentioned above, which are more inclined to one of the classes, AUC gives a clear picture about the ability of the classifier in prediction. The higher the AUC, the higher is the model's capability to distinguish between the classes. From table 6.1, the AUC score computed is 0.772. Computing the ROC curve for the CatBoost classifier is an inbuilt capacity of the CatBoost library of python, and it helps to compute the AUC curve over multiple iterations to obtain the best fit. We use the loss function as "AUC" to get the ROC curve from the classifier as shown in figure 6.4.



Figure 6.4: ROC Curve of the Test Data - Training Flow

• Model Explainability: The model's explainability is shown using SHAP values (12). SHAP values interpret the impact of having a certain value for a given feature in comparison to the prediction we'd make if that feature took some baseline value. Figure 6.5 shows the features and their SHAP values which lead to influencing the predictions of the Classifier.



6.2 Evaluating CatBoost Model Performance - Chosen Performance Metrics

Figure 6.5: CatBoost Explainability of the Test data - Training Flow

6.2.2 Forecast Performance Metrics & Visualizations

Post the classification and with the metrics in the above section, we find the best classifier for our problem. As shown in the figure 5.1, our process of estimating the forecast consists of a mathematical approach on top of the classifier output, which finishes off with visualizations to support the process. To assess both the performance of the final forecast and the visualizations, we formulate two metrics:

• Forecast Value accuracy: The forecast value accuracy is calculated as shown in equation 6.7. The value C is the adjusted forecast value calculated via the mathematical approach mentioned in section 5.3. We also know the actuals sales value for the data present in the test set which is taken as the value for D. We calculated the accuracy of our estimated forecast to be 83% which means our C value only missing from the D value by a margin of 17%. Thus, our results show that the predicted and formulated forecast can be counted as reliable. Although there is a 17% margin, this is estimated in comparison with the actual sales in that time period, which doesn't always need to tally with the forecast and is dependent on the other factors such as the business of the company and has other external factors as well. Nine out of ten times, a good performing company is expected to exceed its forecasted value and there are very few chances for them to be below the forecasted value. This forecast value accuracy is not a metric which will be provided by Evidently, and this is only to prove the performance of the model in order to explain the results of the thesis.

$$C = Calculated forecast value from the model$$
 (6.5)

$$D = Actual \ sales \ happened \ in \ that \ time \ period$$
 (6.6)

Forecast Value accuracy =
$$C/D$$
 (6.7)

• Visualization Benefit: The visualization impact cannot be monitored with the help of any metrics. The visualization can be interpreted in terms of the benefits it brings to analysing and improving the forecast process. The dashboards provide three different information related to Pipeline Quality, Pipeline hygiene and also acts as an output interface to the predictor and mathematical approach's results. The benefit of the visualization is not immediate and rather it takes a long term approach. Over time, these dashboards can be interpreted and could form an integral part of discussions between the sales teams and the management teams to improve the quality of the data in the pipeline and also improve the business standards of the company. When actions are appropriate, it will lead to improvements in the forecasting accuracy and classification accuracy, as these dashboards are a direct reflection of the Sales pipeline data being used.

Infer Dashboard										
		EMEA & LATAM		132.2M		Multi Area				
Forecast	t Value - Current Month	Asia Pacific		65.6M		I Asia Paofic ✓ EMEA & LATAM				
		International		197.8M						
				Detail	ed View - List of I	Pipeline Deals				
Opp Id										
520	ISR-TB-SLO_UNISTA	R_UKC_SSD_UPGRADE - DR-2	144336	EMEA 8 LATAM	EM-Channel Led J	rea Klinichi Center	FAS Hybrid Config	0	Will be Lost	
12272	Nb. FTS. Rehau. SA	P-Shelves_FTS-P.Vigneswar	en #0000765366_DR-2338087	EMEA 8 LATAM	EM-Germeny Ar	ee Reheu AG & Co	FAS Hybrid Centio	28.375	Will be Lost	
					ONTAP Add On	962	Will be Lost			
12949	Lutech x Alstom - Te	ch refrersh x Denimerce - DR-	2341120	EMEA & LATAM	EM-Ent Focus Ar	ea Alatom Ferroviaria SpA	FAS Hybrid Config	79,172	WilbeLost	
22338	Shelferweiterung - D	R-2384328		EMEA & LATAM	EM-Ent Focus Ar	ea SIG Combiblec GmbH & Co KG	FAS Hybrid Config	10,682	Will be Lost	
37585	Para Bilgi Teknolojil	eri - 0PP-01667T		EMEA & LATAM	EM-Channel Led J	rea Para Bilgi Teknolojileri	FAS Hybrid Config	5,000	Will be Lost	
37586	Doga Koleji - OPP-03	667L		EMEA 8 LATAM	EM-Channel Led J	rea Doga Koleji	FAS Hybrid Config	5,000	Will be Lost	
40005	PL_MFG_Kirchhoff_	Modernizacja srodowiska DR.	.KL · DR-2371286	EMEA & LATAM	EM-Channel Led J	rea Kirchhoff Polska Sp.zoo	FAS Hybrid Config	156,864	Will be Lost	
							ONTAP Add-On	131,365	Will be Lost	
43820	HO			EMEA & LATAM	EM-Ent Focus Ar	ea Kapsch BusinessCom AG	NetApp HCI	497,211	Will be Lost	
53255	TECHREFRESH - DR-	2244150		EMEA 8 LATAM	EM-Iberia & LATAV	Area Camara Municipal de Idanha a Nova	FAS Hybrid Config	18,270	Will be Lost	
73317	Destek Merikul AS - I	DPP-016671		EMEA & LATAM	EM-Channel Led J	rea Destek Merikul AS	FAS Hybrid Config	5,000	Will be Lost	
87238	The Value System_S	ENA Development_Red Excel	lent_Storage for ERP - DR-2357987	Asia Pacific	GCASK Area	Sena Development Public Co Ltd.	FAS Hybrid Config	24,000	Will be Lost	
87538 BSc_FTS_Festo-AG_SeeS-Backup-0365FTS_DR00303935		EMEA 8 LATAM	EM-Germany An	ea Festo AG & Co. KG	SAAS Backup	463,203	Will be Lost			
91520 GOV-SK/EHa/MŠ SR/Cyber Security Lab - DR-2371459		71459	EMEA & LATAM	EM-Channel Led J	rea Ministerstvo skolstva SR	FAS Hybrid Config	19,883	Will be Lost		
115322 Prosol_TCE_Bandejas - DR-2406852			EMEA & LATAM	EM-Iberia & LATAN	Area TELEFONICA COMPRAS ELECTONICAS SL	ONTAP Add-On	38,998	Will be Lost		
322195	322195 Atomic Austria GmbH-0PP-322196-Storage NEU, Ablitise HP		EMEA 8 LATAM	EM-Ent Focus Ar	ea Atomic Austria GmbH	High-Performance Flash Config	12,180	Will be Lost		
322532	322532 Kuveyt Turk-OPP-322932-SolidFire_KuveytTurk			EMEA & LATAM	EM-Channel Led J	rea Kuveyt Turk	Solidfire AFA	150,000	Will be Lost	
322957	322967 Mopas Hipermarketler Birligi-OPP-322957-HCI			EMEA & LATAM	EM-Channel Led J	rea Mopas Hipermarketler Birligi	NetApp HCI	75,000	Will be Lost	
329322	DR00029248-Additional DS		EMEA 8 LATAM	EM-Ent Focus Ar	ea Khalifa Fund for Enterprise Development	Renewals	3,702	Will be Lost		
332789	ISR-MS-RO-Alef-ETA	20-Enel Distributie-DR00030	942-New Eseries system	EMEA 8 LATAM	EM-Channel Led J	rea Enel Distributie Muntenia SA	E-Series Hybrid	26,416	Will be Lost	
334135	DR00032146-Pun H	aing Campus_New Storage		Asia Pacific	GCASK Area	Dulwich College	FAS Hybrid Config	30,000	Will be Lost	
364640	DR00033509-VSTEC	S_BCEL_Datacom_DC DR		Asia Pacific	GCASK Area	Banque Pour le Commerce Exterieur Lao Pu blic	FAS Hybrid Config	30,000	Will be Lost	
367772	ABa_FTS-DON_Prog	roup_Backup-FAS(Landau)	InoSYSTEMS_FTS_DR00035848	EMEA 8 LATAM	EM-Germeny An	ea Progroup AG	FAS Hybrid Config	44,343	Will be Lost	
413505	DR00042306-VST_0	IDTH_MONO_DR		Asia Pacific	GCASK Area	Mono Technology Public Company Ltd.	FAS Hybrid Config	25,000	Will be Lost	
414557	Government of Israe	I/Ministry of Defense-OPP-41	4557-MD-KIL: Arbel 135	EMEA 8 LATAM	EM-Ent Focus Ar	ea Government of Israel/Ministry of Defense	FAS Hybrid Config	498,831	Will be Lost	
							Professional Services	1,125	Will be Lost	
495038	DR00050207-Aero Gearbox International - macierz vs VNX		EMEA & LATAM	EM-Channel Led A	rea Aero Gearbox International	FAS Hybrid Config	18,270	Will be Lost		
503004 NNA_PLTA POlitech_Chabas_E2800_DR00054799		EMEA 8 LATAM	EM-Channel Led J	rea Politech SP 200	E-Series Hybrid	18,057	Will be Lost			
501007 DR0005723B-R0 LST TechFresh FMC		EMEA & LATAM	EM-Channel Led J	rea Continental Automotive Products SRL	FAS Hubrid Confin	10.786	Will be Lost			

Figure 6.6: Predictor Output Dashboard

6.2 Evaluating CatBoost Model Performance - Chosen Performance Metrics



Figure 6.7: Pipeline Hygiene - Active



Figure 6.8: Pipeline Health - Past



Figure 6.9: Pipeline Health - Active

6. EXPERIMENT & EVALUATION OF RESULTS

7

Challenges of Machine Learning in B2B Sales Forecasting

With Machine learning on B2B Sales data, we came across some interesting challenges with respect to data handling and classifier performance. In this chapter, we try to list out the challenges faced and explain the approaches we took to fully or partially resolve them. Here, we focus more on B2B Sales Forecasting and the challenges faced over here are applicable to B2B based sales pipeline data. Most of the research in this area doesn't discuss the after effects of the machine learning model, but instead shed light on what kind of business challenges we can expect from the data. This was the main motivation to write this chapter.

7.1 Categorical Encoding Difficulties

The idea behind why we opted to encode the categorical values using CatBoost encoding is explained in section 4.3.3. Additionally, the motivation is also driven by the challenges we faced with categorical encoding of the data in the Active flow. The B2B sales pipeline data used for this thesis is highly categorical, and some features are highly cardinal as well, which were already explained in the previous sections. The B2B Sales pipeline data is constantly changing and although the features remain the same, their values can change over time. During the testing of the active flow, we actually tested two classifiers - namely RandomForest classifier and CatBoost classifier. Although CatBoost was selected in the end, we wanted to test the usage of Random Forest as well since it performed too close with the CatBoost. We designed two implementations of the active flow using the above two classifiers to monitor the performance. On running the two implementations, we found

7. CHALLENGES OF MACHINE LEARNING IN B2B SALES FORECASTING

that the RandomForest Classifier was failing. Upon investigation, we observed the presence of new values for some categories, which the classifier was unable to interpret as it has not seen it during training. The potential areas where we could identify changes in categorical values were

- Sales Team There were new sales teams created, and some teams were split into two with newer names to suit the changes in the business model. This led to the active flow receiving these values, since the data for the active flow is new and fresh.
- Products New portfolio of products or renaming the existing products. The effect of renaming the existing products was minimized by mapping the new names with the existing names in the data preprocessing stage.
- Areas and Regions Changing the business model or upgrading it obviously has an impact on how the mapping of areas and regions are made. Like products, there were newer areas and some existing areas were either grouped into another one or split into different ones.

This kind of difficulties are more common in a B2B sales pipeline as companies evaluate and upgrade their business model and portfolios every year to be in line with the evolving market conditions and also to improve their business understanding. We resolved this issue by using the CatBoost encoding which has inbuilt capacity to account for unseen values by using its encoding methodology as explained in the section 4.3.4. The CatBoost classifier using the CatBoost encoding scheme (15) was able to successfully run and produce the expected results in spite of encountering new data. One can also wonder that to counter the effect of the splitting up of already existing feature values into newer values (for example: North Holland into Amsterdam and Amstelveen) could be offset by retraining the machine learning model and is not a categorical encoding problem. Yes, that is a possibility, but we do have some issues on the data side to successfully apply the method, which will be explained in the later sections. This problem of categorical encoding also gives rise to the problem of Data Drift.

7.2 Data Drift

Models which deal with streaming data experience the problem of data drift. The production models are faced with new data, and the distribution of the data can change over time due to a variety of factors. Data drift is the change in the distribution of the baseline dataset on which the model was trained and the current real time production data. In our case, we evaluate the data drift with the help of Evidently AI's data drift reporting. As shown in the 5.4 we have a model performance classification block that monitors the production model on a scheduled basis and generates performance reports, and it also provides us with the Data drift report. Classification accuracy of Machine learning models generally have the ability to deteriorate over time with changes in data, which is termed as Model drift. Model drift is primarily caused due to two factors - Concept Drift and Data Drift. Concept drift is the phenomenon that happens when the meaning of what you are trying to predict changes. Concept drift is concerned with the changes in the prediction of the target variable.(29)(30)(31) For example, a corona prediction model designed to predict whether a person will have corona or not would have experienced concept drift. This is because, in the beginning only few symptoms of corona were known, and the model predicts based on those but with the increase in research we have identified newer symptoms that are unknown to the model. The concept has changed over time and if we don't retrain the model, then it is going to perform badly. The motivation to looking into this aspect of production based machine learning models stemmed from our model's drop in classification levels when connected to the active flow. We experienced a drop of 6%initially in our classification accuracy from our estimated classification accuracy of 77% on the test dataset of the training flow.

Upon investigation of the data drift report generated from the model by comparing the data from the training pipeline and the fresh incoming data from the Active pipeline, we formulated the following points which could have been producing the data drift,

- Week :- The week feature consists of one extra week for leap years. Our training data doesn't contain data for leap years. For leap years, we have 53 weeks compared to the normal 52 weeks. This means that there is one week of data that is unseen by the model, and the distribution of the week is not present in the training dataset.
- More Samples:- In some instances, the incoming data on the active flow had a higher distribution for particular features. For example, only five samples for a particular category in the training data, but there are more samples in the incoming data. This can lead to concept drift, where newer combinations are coming in which the model has not yet trained on.
- New Samples:- As stated in the previous section on the category encoding challenges, newer values can come in which are not a part of the training set, leading to changes

7. CHALLENGES OF MACHINE LEARNING IN B2B SALES FORECASTING

in distribution. This differs from the previous point because here we have entirely new values, whereas in the previous one, we have the values, but newer combinations could be potentially made.

The figures 7.1 and 7.2 shows some features and their interesting data drift charts.



Figure 7.1: Drift - Sales Area



Figure 7.2: Drift - Duration

With that, we tried working around some features to improve the production model. For instance, we dropped the week level feature since it was a great cause, and we also dropped a similar feature which was representing the segmentation of customers as per the area in which they operate. With these, we were able to improve the model to predict at a 75% of accuracy, which is four points from the original state. We cannot classify it as a resolved issue because this is currently the case and this can change in the future as some other features can drift from the trained sample. This is mostly a problem when you use categorical data like the ones present in the B2B sales dataset. Some of the data challenges we had are explained in the next section.

7.3 Data Challenges

As the B2B business process keep changing over time, the data keeps changing as well. We list out potential data challenges which was interesting to understand when building the model.

- The ability of the entire data structure to change itself. Mostly B2B Companies group data based on the sales territory, customer segmentation and product grouping hierarchies. This is a periodic change in the companies with change in the business scheme of things. For example, today you have 59,000 records grouped under a geography like Europe. Tomorrow the company implements a new plan where due to business needs they are going to split the geography of Europe into Europe-West and Europe-East & Central. This would potentially mean the data structure is changing, and the machine learning models needs to be retrained and figured out again.
- Data Quality: As mentioned across the thesis, data quality is a major issue in formulating correct forecasts or predictions.
- Reinstatements of Data as per new structure takes longer time in a company as years of historical data needs to be aligned into the newer format, which potentially hinders the machine learning approach's capability to predict correctly. This is because with longer wait times, the model is still trained on the old set of data, which would give rise to the other problems mentioned in the previous sections of this chapter.

7. CHALLENGES OF MACHINE LEARNING IN B2B SALES FORECASTING

8

Limitations and Future work

The thesis gives an alternative approach to formulate the forecast in a B2B Sales setup. Although we were able to successfully devise a strategy to formulate the forecasts using the Sales pipeline data, there are some drawbacks which we don't fully overcome. This chapter explains in brief the limitations of the proposed framework and also talks about the future work which can be done to improve the forecasting process using machine learning.

8.1 Limitation

Every approach has its own advantages and limitations. It is impossible to design a framework or a working methodology without any limitations attached to it. In this regard, our framework has three specific limitations. Firstly, the model is seen as an alternative approach to use machine learning and arrive at a forecast which will challenge the conventional forecasting methodology. The problem stems from the adaptability of the sales professionals and the sales teams within the organization. We conducted an initial one on one discussions with a few members from different sales teams present in the organization to understand the likeability of this data driven approach. The outcome of the discussions was that although they were accepting the level of estimates the model produced, yet they were not so much in favour of considering it as a challenger. Organizations are also not fully interested to drive away from the conventional forecasting process or look at an alternative to challenge the forecast methodologies. Thus, we find very few related literature in this area when compared to sales forecasting in B2C domains where machine learning is becoming widely accepted.

The second limitation is the mathematical approach present in our framework. This works for most of the scenarios to churn out a stable forecast, but the mathematical

8. LIMITATIONS AND FUTURE WORK

approach has no use if the sales value associated with an opportunity is entirely correct. This is an ideal scenario, but the mathematical approach can be discarded even if we have ninety percent of opportunities with correct values, as it will not increase or decrease the forecast value to induce the problems of under/over forecasting. The same data is used to formulate the conventional forecasts, but the sales professionals manually ensure that the correct sales value are recorded for each opportunity before using these opportunities for forecasting. This is also done on offline Excel files, and it's not corrected back in the CRM system from which we pull the data.

The third limitation is that at this moment we predict based on open and closed opportunities. Related work in this field also shows some approaches where they consider the opportunities which are in the other sales stages as shown in 2.1 and mark them as active and predict the same. When one looks up the historical sales pipeline data, There are very few opportunities which will be in the active state from the past. But this scenario is found in abundance with opportunities that are in the current or future sales pipeline. Although, our model learns the won/lost behaviour of the opportunities and uses them to predict the new deals, it's still a valid scenario to look into how to predict the different stages of the opportunities instead of just predicting the final stage which can help organizations to formulate other risk measures related to forecast and build more trust around the data driven model.

The last limitation which we would like to talk about is the ability of the external factors to influence the forecast processes. Although we don't see any drop in the sales behaviour of the organization due to pandemic, which helped to build a stable model, it is still not so easy to discard. The pandemic caused organizations to back out of their plans or downsize, leading to many significant cut down in revenue. Since B2B Sales is based on business to business, these factors will have a huge impact in estimating the forecast for a time period. Also, there are organization based issues (32) which can make a business back out of from its promise of buying, social and demographic factors like change of policies, governmental changes of an area etc. which can potentially impact the forecast. The data driven approach doesn't take into consideration factors like these while formulating the forecast, which makes it a less likely solution at times when these factors play a role. With this we provide an overview of the limitations and although there can be many more small limitations associated with the approach.
8.2 Idea - All stage B2B Predictor

As a future work, we would like to propose an idea before we jump into the conclusion part of the thesis. To approach the third limitation stated in the final section 8.1, we propose to take a different approach in terms of data storage, which can elevate the forecasting approaches. At the moment, each opportunity is represented with the means of a unique row in the dataset and the different stages of the opportunities are not recorded individually, but instead the sales professionals update the stage of the opportunity to the current stage. This leaves us with no ability to track the different stages the opportunity went through before finally going into the won/lost stage. As mentioned in the Background chapter of this paper, it is not a golden rule that the opportunity needs to go through every sales stage in order to get closed. It can directly go to the closed stage from the initial stage, or it can traverse some stages before it goes into the final stage. Keeping all records related to an opportunity pertaining to different sales stages is a mundane task and will result in the requirement of increased storage spaces and large volumes of data. So what we propose is to add another dimension to the data, where we have separate indicator features for each sales stage. A "Yes" will denote that the opportunity passed through this stage, and "No" will denote the opportunity did not go through that stage. With this we can identify the behaviour of the opportunity in terms of traversing the sales stage which will give a comprehensive picture of the progress an opportunity makes. This will also enhance the classification metric as the model can learn from the behaviour whether the opportunity will make a traversal across different stages or just abruptly enter into the omitted or lost stages. With this, we can also somewhat positively influence the final limitation mentioned in the section 8.1 which states the unaccountability of external factors. By noting the traversal of the stages, we can account for the buyer based factors such as backing out of a promise, internal problems etc as these deals don't traverse different stages, but they are just marked as omitted and are not considered for forecasting.

Although this idea seems like a change to the normal approach, the ability to learn about the opportunity's traversal of different stages will bring added benefits to the forecasting in a data driven approach.

8. LIMITATIONS AND FUTURE WORK

9

Conclusion

Finally, we arrive at the last section of the thesis. With B2B Sales forecasting as the area of research for the thesis, we designed a model that consisted of three different components that together work to achieve the goal of the thesis, which was to use a data driven method to predict the won/lost nature of the opportunities and estimate a forecast value. The evaluations of the results from the model were promising and helps to provide an alternative for the conventional forecasting methodology followed in companies. Additionally, it gives the ability to make data driven decisions instead of having only the results from the conventional forecasting processes that are influenced by decisioning of the individual or the team. We list the research questions formulated for the thesis and state the conclusion we got from them.

1. RQ1.0 - Can a Sales Forecast formulated using Machine Learning challenge the Sales Forecast generated by the Knowledge of the sales team in terms of Quality?

The model formed by using the three components mentioned provides an alternative data driven approach which can be used to challenge the conventional forecasting processes of the sales team to an extent. With the quality of the data we have at hand, we were able to produce a 77% classification accuracy on predicting the opportunities and the forecast value formulated using both the classifier and the mathematical approach was found to be 83% accurate to the actual sales happened in the same time period. With improvements in data quality as mentioned in the above chapters, the model can be improved even further.

• Which Machine Learning Algorithm performs better with respect to the prediction? From analysing the classification metrics we took to evaluate the different classifiers, we found out CatBoost was found to be the best model for carrying out the predictions.

• How expert's knowledge can be incorporated into modelling the data?

The expert's knowledge was included in many sections of this model. It was included to carry out discussions to find out about the adaptability of the model. Expert's knowledge was instrumental in developing an understanding of the data. It also helped in feature selection, as we consider some features as part of our model which are not found to be important by the feature selection approaches. The knowledge was also instrumental in formulating the mathematical approach to achieve the forecast values and also in evaluating the results of the model. So the expert's knowledge was an integral part in bringing this model to life.

• How the model results can be evaluated?

We use different classification metrics to evaluate our model such as the AUC-ROC curve, F1 Score and the classification accuracy. We also design a metric to calculate the accuracy of the mathematical approach, which are discussed in the experiment & evaluation of results chapter.

• What level of accuracy is accepted for the model to be deployed commercially?

From related work, we could see higher accuracies obtained for the similar problems. Keeping the quality of the data at hand, we aimed for a 75% accurate forecast on predictions and the estimated forecast value to be at least 80% accurate. We are happy to see that the model's performance exceeded these accuracies.

• RQ2.0- How Data Drift plays a role in B2B Sales Data? Does it have a significant impact, and What can be done to minimize the changes in model performance, if any?

We faced different challenges when coming up against the challenges of machine learning in B2B Sales forecasting. Difficulties in categorical encoding, Data challenges related to B2B Sales Data and data drift. We observed data drift on the active flow of the model, which uses the new incoming data into the system as the input. The model performance deteriorated when handling new data from 77% to 71%. With Evidently AI, we formulated the data drift report, which helped us analyse which features contributed to this phenomenon. Based on the drift report, we made removed some features which helped improve model performance back to 75% in terms of classification accuracy. This data drift report was also monitored and formed a part of the active flow to identify if the model needs to be retrained or not and to keep a check on the distribution of the incoming data.

1. What are the KPI's with respect to Sales Pipeline/Predictor Output and does using Visualizations a good way to understand the Sales Pipeline?

We initially wanted to look at formulating KPI's with which we designed this research question, but this changed during the course of the thesis. As there were multiple factors, designing KPI's for each of them would have resulted in an exodus of KPI's which would have not helped the scope of the thesis. The idea was altered to provide visualizations which can bring discussions about the Sales pipeline and also help improve the quality of the data flowing into the system. Also, visualizations were used to convey the results from the forecasting process. Thus, with this we formulated the visualizations to cover the three topics - Sales Hygiene, Sales Health and the output of the forecasting process. This provides an understanding of the sales pipeline and also helps to achieve a learning about the sales pipeline and provides a way to improve them.

• What kind of Visualizations can be used for this process?

Four separate dashboards namely Predictor Output dashboard, Pipeline Hygiene - Active, Pipeline Health - Active/ Past were created consisting of different types of visualizations to support the three areas we intended to cover. The individual visualizations present in the dashboards are discussed in chapter 5.

Results from this thesis can be used to add some significant research into the application of Machine learning techniques in the area of B2B Sales Forecasting. This also provides a way for organizations to apply this and start using and trusting machine learning in this area. While we have provided a work which provides acceptable results, this is still an area where a lot can be done. More work needs to be done by companies in terms of recording

9. CONCLUSION

different dimensions of data and improving the quality of the data in the Sales pipeline. Due to limited resources, we couldn't perform some machine learning side techniques to improve the classifier's performance like grid search to find optimal parameters as they were computationally expensive with increased data size. Also, with the evolving technologies, sales professionals should be encouraged to develop an understanding of machine learning and advanced techniques to enable researchers to explain about the benefits of them. This will help improve adaptability. Furthermore, as a conclusion, neural networks and deep learning models are not used in this area due to their issues with explainability. Work needs to be done to improve the explainability of these models, which can provide a lift to using data driven approaches in this area. Apart from these, machine learning still works as a good alternative to conventional forecasting techniques to formulate the forecasts with which we would like to conclude this thesis.

References

- MARKO BOHANEC, MIRJANA KLJAJIC BORSTNAR, AND MARKO ROBNIK-SIKONJA. Integration of machine learning insights into organizational learning: A case of B2B sales forecasting. In 28th Bled eConference: s#eWellBeing, Bled, Slovenia, June 7-10, 2015, page 33, 2015. 13
- [2] JUNCHI YAN, MIN GONG, CHANGHUA SUN, JIN HUANG, AND STEPHEN M. CHU. Sales pipeline win propensity prediction: A regression approach. In REMI BADONNEL, JIN XIAO, SHINGO ATA, FILIP DE TURCK, VOICU GROZA, AND CAR-LOS RANIERY PAULA DOS SANTOS, editors, IFIP/IEEE International Symposium on Integrated Network Management, IM 2015, Ottawa, ON, Canada, 11-15 May, 2015, pages 854–857. IEEE, 2015. 14
- [3] STEPHEN MORTENSEN, MICHAEL CHRISTISON, BOCHAO LI, AILUN ZHU, AND RA-JKUMAR VENKATESAN. Predicting and Defining B2B Sales Success with Machine Learning. In 2019 Systems and Information Engineering Design Symposium (SIEDS), pages 1–5, 2019. 14
- [4] LIOR ROKACH AND ODED MAIMON. Decision Trees, 6, pages 165–192. 01 2005. 14
- [5] LEO BREIMAN. Random Forests. Machine Learning, 45(1):5–32, 2001. 14
- [6] TIANQI CHEN AND CARLOS GUESTRIN. XGBoost: A Scalable Tree Boosting System. In BALAJI KRISHNAPURAM, MOHAK SHAH, ALEXANDER J. SMOLA, CHARU C. AGGARWAL, DOU SHEN, AND RAJEEV RASTOGI, editors, Proceedings of the 22nd ACM SIGKDD International Conference on Knowledge Discovery and Data Mining, San Francisco, CA, USA, August 13-17, 2016, pages 785–794. ACM, 2016. 14
- [7] BRENDAN ANDREW DUNCAN AND CHARLES PETER ELKAN. Probabilistic Modeling of a Sales Funnel to Prioritize Leads. In Longbing Cao, Chengqi

REFERENCES

ZHANG, THORSTEN JOACHIMS, GEOFFREY I. WEBB, DRAGOS D. MARGINEANTU, AND GRAHAM WILLIAMS, editors, *Proceedings of the 21th ACM SIGKDD International Conference on Knowledge Discovery and Data Mining, Sydney, NSW, Australia, August 10-13, 2015*, pages 1751–1758. ACM, 2015. 14

- [8] ALIREZA REZAZADEH. A Generalized Flow for B2B Sales Predictive Modeling: An Azure Machine Learning Approach. CoRR, abs/2002.01441, 2020.
 15
- [9] GUOLIN KE, QI MENG, THOMAS FINLEY, TAIFENG WANG, WEI CHEN, WEIDONG MA, QIWEI YE, AND TIE-YAN LIU. LightGBM: A Highly Efficient Gradient Boosting Decision Tree. In ISABELLE GUYON, ULRIKE VON LUXBURG, SAMY BENGIO, HANNA M. WALLACH, ROB FERGUS, S. V. N. VISHWANATHAN, AND ROMAN GARNETT, editors, Advances in Neural Information Processing Systems 30: Annual Conference on Neural Information Processing Systems 2017, December 4-9, 2017, Long Beach, CA, USA, pages 3146–3154, 2017. 15
- [10] XIN XU LEI AND TANG VENKAT RANGAN. Hitting your number or not? A robust & intelligent sales forecast system. In JIAN-YUN NIE, ZO-RAN OBRADOVIC, TOYOTARO SUZUMURA, RUMI GHOSH, RAGHUNATH NAMBIAR, CHONGGANG WANG, HUI ZANG, RICARDO BAEZA-YATES, XIAOHUA HU, JEREMY KEPNER, ALFREDO CUZZOCREA, JIAN TANG, AND MASASHI TOYODA, editors, 2017 IEEE International Conference on Big Data, BigData 2017, Boston, MA, USA, December 11-14, 2017, pages 3613–3622. IEEE Computer Society, 2017. 15
- [11] TIEMO THIESS, OLIVER MÜLLER, AND LORENZO TONELLI. Design Principles for Explainable Sales Win-Propensity Prediction Systems. In NORBERT GRONAU, MOREEN HEINE, HANNA KRASNOVA, AND K. POUSTCCHI, editors, Entwicklungen, Chancen und Herausforderungen der Digitalisierung: Proceedings der 15. Internationalen Tagung Wirtschaftsinformatik, WI 2020, Potsdam, Germany, March 9-11, 2020. Zentrale Tracks, pages 326–340. GITO Verlag, 2020. 15
- [12] SCOTT M. LUNDBERG AND SU-IN LEE. A Unified Approach to Interpreting Model Predictions. In ISABELLE GUYON, ULRIKE VON LUXBURG, SAMY BENGIO, HANNA M. WALLACH, ROB FERGUS, S. V. N. VISHWANATHAN, AND ROMAN GAR-NETT, editors, Advances in Neural Information Processing Systems 30: Annual Conference on Neural Information Processing Systems 2017, December 4-9, 2017, Long Beach, CA, USA, pages 4765–4774, 2017. 16, 46

- [13] KEDAR POTDAR, TAHER S. PARDAWALA, AND CHINMAY D. PAI. A Comparative Study of Categorical Variable Encoding Techniques for Neural Network Classifiers. International Journal of Computer Applications, 175:7–9, 2017. 22
- [14] PATRICIO CERDA AND GAËL VAROQUAUX. Encoding high-cardinality string categorical variables. CoRR, abs/1907.01860, 2019. 22
- [15] LIUDMILA OSTROUMOVA PROKHORENKOVA, GLEB GUSEV, ALEKSANDR VOROBEV, ANNA VERONIKA DOROGUSH, AND ANDREY GULIN. CatBoost: unbiased boosting with categorical features. In SAMY BENGIO, HANNA M. WALLACH, HUGO LAROCHELLE, KRISTEN GRAUMAN, NICOLÒ CESA-BIANCHI, AND ROMAN GAR-NETT, editors, Advances in Neural Information Processing Systems 31: Annual Conference on Neural Information Processing Systems 2018, NeurIPS 2018, December 3-8, 2018, Montréal, Canada, pages 6639–6649, 2018. 23, 32, 52
- [16] ANNA VERONIKA DOROGUSH, VASILY ERSHOV, AND ANDREY GULIN. CatBoost: gradient boosting with categorical features support. CoRR, abs/1810.11363, 2018. 23, 32
- [17] ESSAM AL DAOUD. Comparison between XGBoost, LightGBM and Cat-Boost Using a Home Credit Dataset. International Journal of Computer and Information Engineering, 13(1):6 – 10, 2019. 23
- [18] GIRISH CHANDRASHEKAR AND FERAT SAHIN. A survey on feature selection methods. Comput. Electr. Eng., 40(1):16–28, 2014. 24
- [19] S. VISALAKSHI AND V. RADHA. A literature review of feature selection techniques and applications: Review of feature selection in data mining. In 2014 IEEE International Conference on Computational Intelligence and Computing Research, pages 1–6, 2014. 24
- [20] MIRON B. KURSA, ALEKSANDER JANKOWSKI, AND WITOLD R. RUDNICKI. Boruta
 A System for Feature Selection. Fundam. Informaticae, 101(4):271–285, 2010.
 24
- [21] DANIELA GOTSEVA ROUMEN TRIFONOV AND VASIL ANGELOV. Binary classification algorithms. International Journal of Development Research, 07, 11 2017. 30, 31

REFERENCES

- [22] ZHI-HUA ZHOU. Ensemble Learning. In STAN Z. LI AND ANIL K. JAIN, editors, Encyclopedia of Biometrics, Second Edition, pages 411–416. Springer US, 2015. 31
- [23] CANDICE BENTÉJAC, ANNA CSÖRGO, AND GONZALO MARTÍNEZ-MUÑOZ. A comparative analysis of gradient boosting algorithms. Artif. Intell. Rev., 54(3):1937–1967, 2021. 31
- [24] DANIEL BERRAR. Cross-Validation. 01 2018. 41
- [25] Catboost Cross validation. 42
- [26] Evidently-AI. 42
- [27] GUY HANDELMA, HONG KOK, RONIL CHANDRA, AMIR RAZAVI, SHIWEI HUANG, MARK BROOKS, MARLON LEE, AND HAMED ASADI. Peering Into the Black Box of Artificial Intelligence: Evaluation Metrics of Machine Learning Methods. *American Journal of Roentgenology*, 212:1–6, 10 2018. 43, 46
- [28] YANGGUANG LIU, YANGMING ZHOU, SHITING WEN, AND CHAOGANG TANG. A Strategy on Selecting Performance Metrics for Classifier Evaluation. International Journal of Mobile Computing and Multimedia Communications, 6:20–35, 10 2014. 44
- [29] GEOFFREY I. WEBB, ROY HYDE, HONG CAO, HAI-LONG NGUYEN, AND FRANÇOIS PETITJEAN. Characterizing concept drift. Data Min. Knowl. Discov., 30(4):964– 994, 2016. 53
- [30] JIE LU, ANJIN LIU, FAN DONG, FENG GU, JOÃO GAMA, AND GUANGQUAN ZHANG. Learning under Concept Drift: A Review. CoRR, abs/2004.05785, 2020. 53
- [31] Understanding and Handling Data and Concept Drift. 53
- [32] DONNA DAVIS AND JOHN MENTZER. Organizational factors in sales forecasting management. International Journal of Forecasting, 23:475–495, 07 2007. 58